Author's response to reviews

Title: Malaria Prevalence and Mosquito Net Coverage in Oromia and SNNPR Regions of Ethiopia

Authors:

Estifanos B Shargie (estifanos_b@yahoo.com)
Teshome Gebre (global2000@ethionet.et)
Jeremiah Ngondi (jn250@cam.ac.uk)
Patricia M Graves (cem4@cdc.gov)
Aryc W Mosher (anm7@cdc.gov)
Paul M Emerson (paul.emerson@emory.edu)
Yeshewamebrat Ejigsemah (yeshe2005@yahoo.com)
Tekola Endeshaw (teko1960@yahoo.ca)
Dereje Olana (dolana@usaid.gov)
Asrat WeldeMeskel (asratwm@yahoo.com)
Admas Teferra (admastr@et.afro.who.int)
Zerihun Tadesse (zerihtad@yahoo.com)
Abate Tilahun (abate_tilahun@yahoo.com)
Gedeon Yohannes (tegicho6@yahoo.com)
Frank O Richards Jr (fxr1@cdc.gov)

Version: 3 Date: 12 September 2008

Author's response to reviews: see over
Dear Editorial Production Team,

Thank you very much for your e-mail of 05 September 2008 and for agreeing to publish our manuscript in BMC Public Health. As per the request of the editorial team, we have now made formatting changes to the manuscript. Below, we have provided a point-by-point response to your comments.

Sincerely,

The authors

Major Revisions

References - References should be cited in the text using consecutive numbers in square brackets. Please do not use superscript numbers. Also, please place the reference before the full stop. For example `text [1].

Response: We have now changed the formatting accordingly.

Figures in text: Figures are also included correctly. Please delete figures found in text section.

Response: Figures in the text section are now deleted and uploaded separately.

Minor Revisions

References- Reference XX has not extracted the authors names correctly. Commas need to be inserted between the surname and forename initial. When that's done it will extract correctly, for example Lewis B; Coates R will extract as L B; C R If you change that to: Lewis, B; Coates, R Then it will extract properly.
Response: Thanks for the reminder. We have extracted and checked the references during the revision uploading process, and the authors names are extracted correctly.

Figure cropping - It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image.

Response: We have cropped the figures to the best possible level.

Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions as this is the final proofing stage for your manuscript, and you will not be able to make changes after acceptance.

Response: Thanks again. We have tried to proofread the manuscript and have now uploaded the revised version.